Author's response to reviews

Title: Completeness and timeliness of Salmonella notifications in Ireland in 2008: a cross sectional study

Authors:

Nathalie Nicolay (nathalienicolay@yahoo.fr)
Patricia Garvey (patricia.garvey@hse.ie)
Niall DeLappe (niall.delappe@hse.ie)
Martin Cormican (martin.cormican@hse.ie)
Paul McKeown (paulj.mckeown@hse.ie)

Version: 5 Date: 14 September 2010

Author's response to reviews: see over
Dear Editors

We would like to thank you for comments made on the editing part. We hope the current answers and the amendments made in the manuscript will satisfy you.

We certify that all the authors read and approved the current manuscript. In addition, we answered specifically each of your comments below.

1. We made sure that the abstract is identical in the manuscript and the submission system.

2. We indicated in the author's contribution section that all the authors read and approved the final manuscript.

3. We updated the position of references 5-9 that were moved to 2-6 and consequently references 2-4 were moved to 7-9.

4. “Introduction” was renamed “Background”

5. Tracks changes were removed.

6. Title “Material and methods” was changed to methods.

7. Citation 10 is now fully provided.

8. Changes were made in table text. Titles were left above the tables and legends were put below the tables.

9. Figures are in the correct orientation.

10. We found other typo errors that we corrected.

Thank you for your comments.

Pr Martin Cormican